Author's response to reviews

Title: Results at seven years after the use of Intracameral cefazolin as a prophylaxis of endophthalmitis in cataract surgery

Authors:

Pedro Romero-Aroca (romeropere@gmail.com)
Isabel Mendez-Marin (imendez@grupsagessa.com)
Merce Salvat-Serra (msalvats@grupsagessa.com)
Joan Fernandez-Ballart (jfernandez@grupsagessa.com)
Matias Almena-Garcia (malmenag@hotmail.com)
Javier Reyes-Torres (jreyes@grupsagessa.com)

Version: 8 Date: 23 January 2012

Author's response to reviews:

Dear Editor

I made the changes suggested in your revision:

1. The competing interest has been eliminated in front page
2. The vertical lines in tables has been eliminated

Sincerely

Pedro Romero-Aroca